Author's response to reviews

Title: Dual exon skipping in myostatin and dystrophin for Duchenne muscular dystrophy

Authors:

Dwi U Kemaladewi (d.u.kemaladewi@lumc.nl)
Willem MH Hoogaars (w.m.h.hoogaars@lumc.nl)
Sandra H van Heiningen (s.h.van_heiningen@lumc.nl)
Samuel Terlouw (samuelterlouw@gmail.com)
David JJ de Gorter (daviddegorter@uni-muenster.de)
Johan T den Dunnen (ddunnen@humgen.nl)
Gert Jan B van Ommen (g.j.b.van_ommen@lumc.nl)
Annemieke Aartsma-Rus (a.m.aartsma-rus@lumc.nl)
Peter ten Dijke (p.ten_dijke@lumc.nl)
Peter AC 't Hoen (p.a.c._t_hoen@lumc.nl)

Version: 7 Date: 12 April 2011

Author's response to reviews: see over
Leiden, April 12, 2011

Subject: Proof

Dear BMC Medical Genomics editors,

We have revised the requested formatting on the manuscript titled ‘Dual exon skipping in myostatin and dystrophin for Duchenne muscular dystrophy’ as follows:

1. Authors affiliation
2. Removal of visible lines in the table

In addition, we did not see the additional figure 1 in the previous links we received. Please take this additional figure into your consideration.

Should you need any further information regarding the manuscript, please let me know.

Yours sincerely,

Dwi Utami Kemaladewi
Center for Human and Clinical Genetics
Leiden University Medical Center
Einthovenweg 20, 2333 ZC Leiden, the Netherlands.
Tel: +31-71-5269423
Fax: +31-71-5268285
E-mail: d.u.kemaladewi@lumc.nl